Author’s response to reviews

Title: A modified Delphi study to identify the features of high quality measurement plans for healthcare improvement projects

Authors:

Thomas Woodcock (thomas.woodcock99@imperial.ac.uk)
Yewande Adeleke (yewande.adeleke13@imperial.ac.uk)
Christine Goeschel (chris.a.goeschel@medstar.net)
Peter Pronovost (peter.pronovost@uhhospitals.org)
Mary Dixon-Woods (director@thisinstitute.cam.ac.uk)

Version: 3 Date: 10 Dec 2019

Author’s response to reviews:

Thank you for highlighting these two issues. 1. The funding and author's contributions sections have been amended to clarify that the funders had no role in design, data collection, analysis, interpretation or writing. 2. All persons acknowledged in the acknowledgements section have given permission to be acknowledged by name.